CSE 326: Data Structures

Win ter 1998

Course Information

Instructor: Martin T ompa Sieg 426B tompa@cs.w ashington.edu 543-92 63

T A: Matthew Ric hardson Sieg 326 mattr@cs.w ashington.edu

O�ce hours: Will b e announced shortly .

Prerequisites: CSE 143, CSE 321.

Grading: Homew ork (40%), Midterm (20%), Final (40%). P ercen tag es are appro ximate. The

�nal exam is sc heduled for Monda y , Marc h 16, 1998, 2:30- 4:2 0 p.m.

Computing facilities: Sieg 329 Unix lab. Programming assignmen ts will b e done in C++.

Course W eb: All handouts and a log of all mail sen t to the class mailing list will also b e a v ai lable

on the course w eb at the follo wing URL:

http://www.cs.washington.edu/education/courses/326/CurrentQtr/

Mailing List: The class mailing list is cse326@cs.washington.edu. Instructions for subscribing

to the class mailing list can b e found on the course w eb.

W e will use this list for announcemen ts of general in terest to the class. Studen ts should also

feel free to use it to ask questions, p ost information, or initiate discussions of general in terest to

the class. Questions or commen ts that are not of general in terest should instead b e directed to the

T A or instructor at the addresses ab o v e.

T extb o ok: Harry Lewis and Larry Denen b erg, Data Structur es & Their A lgorithms, Harp er-

Collins, 1991.

Reading assignmen t: P ages 1{25, b y next lecture. The sections on \Notati on for Programs",

\Lo cativ es", \Abstract Data T yp es", and \Order Notatio n" are particularly cen tral to this course.

If y ou w ould lik e to request academic accomo dations due to a disabilit y , please con tact Disabled

Studen t Services, 448 Sc hmitz, 543{8 92 4 (V/TDD) . If y ou already ha v e a letter from Disabled

Studen t Services indicating y ou ha v e a disabilit y that requires academic accomo dati ons, please let

the instructor kno w so w e can discuss the appropriate accomo datio ns.